Author's response to reviews

Title: MicroRNA-21 and the Clinical Outcomes of Various Carcinomas A Systematic Review and Meta-Analysis

Authors:

Wenjia Wang (alostfish@163.com)
Jinhui Li (ielts456@163.com)
Wei Zhu (zhuyc126@126.com)
Chen Gao (gaochen.science@gmail.com)
RuiJingfang Jiang (ruijingfang.jiang@stud.ki.se)
Wenxue Li (liwenxue_2000@163.com)
Qiansheng Hu (huqsh@mail.sysu.edu.cn)
Bo Zhang (zhangb65@mail.sysu.edu.cn)

Version: 2 Date: 3 July 2014

Author's response to reviews:

1. provide Line numbering
2. provide Acknowledgements section.:as our research didn't gotten any funds, we add "none" in our manuscript.
3. Ethics statement: as we found some of the meta-analysis which have published in your magazine didn't give this kind of information, our manuscript of this meta-analysis didn't show this, too. Thanks!